

Temi di Discussione

(Working Papers)

Two EGARCH models and one fat tail

by Michele Caivano and Andrew Harvey







Temi di discussione

(Working papers)

Two EGARCH models and one fat tail

by Michele Caivano and Andrew Harvey

Number 954 - March 2014

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Giuseppe Ferrero, Pietro Tommasino, Margherita Bottero, Giuseppe Cappelletti, Francesco D'Amuri, Stefano Federico, Alessandro Notarpietro, Roberto Piazza, Concetta Rondinelli, Martino Tasso, Giordano Zevi. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

TWO EGARCH MODELS AND ONE FAT TAIL

by Michele Caivano* and Andrew Harvey†

Abstract

We compare two EGARCH models, which belong to a new class of models in which the dynamics are driven by the score of the conditional distribution of the observations. Models of this kind are called dynamic conditional score (DCS) models and their form facilitates the development of a comprehensive and relatively straightforward theory for the asymptotic distribution of the maximum likelihood estimator. The EGB2 distribution is light-tailed, but with a higher kurtosis than the normal distribution. Hence it is complementary to the fat-tailed t. The EGB2-EGARCH model gives a good fit to many exchange rate return series, prompting an investigation into the misleading conclusions liable to be drawn from tail index estimates.

JEL Classification: C22, G17.

Keywords: exchange rates, heavy tails, Hill's estimator, score, robustness, EGB2, Student's t, tail index.

Contents

1.	Introduction	5
2.	Tails and tail indices	7
3.	Dynamic conditional score volatility models	9
	3.1 Maximum likelihood estimation	. 11
	3.2 Beta-t-EGARCH	. 13
	3.3 Gamma-GED-EGARCH	14
4.	EGB2-EGARCH	. 14
	4.1 Properties of EGB2	. 15
	4.2 Dynamic scale model	. 17
	4.3 Maximum likelihood estimation	. 20
5.	Exchange rates	22
6.	Scale parameters and tail indices	24
	6.1 Tail index estimators	25
	6.2 Tail index estimators of shape parameters for EGB2 and Student t distributions	26
	6.3 Estimates of shape parameters from tail indices of residuals	31
	6.4 Tail index estimates for raw data	. 35
7.	Conclusions and extensions	. 36
Re	eferences	. 39

^{*} Bank of Italy, Economic Outlook and Monetary Policy Directorate; e-mail: michele.caivano@bancaditalia.it.
† University of Cambridge, Faculty of Economics; e-mail: ach34@cam.ac.uk.

1 Introduction

The exponential generalized autoregressive heteroskedasticity (EGARCH) class of models was introduced by Nelson (1991) as a means of modeling changing volatility. By letting the dynamic equation for the logarithm of variance be driven by the score of the conditional distribution, many of the theoretical problems inherent in EGARCH models are resolved; see Harvey (2013, ch 4). Furthermore there is already a body of evidence showing that these dynamic conditional score (DCS) EGARCH models perform better on real data than do standard GARCH formulations; see, for example, Creal et al (2011) and Harvey and Sucarrat (2012).

The DCS-EGARCH model based on a conditional Student-t distribution, called Beta-t-EGARCH, is resistant to observations that would be outliers if a Gaussian distribution were used. The reason for this robustness is that the score depends on a beta variable which, of course, is bounded; compare the discussion of robust GARCH models in Muler and Yohai (2008). The t-distribution has fat tails (for finite degrees of freedom) and this property is reflected in the shape of the score function. However, not all variables which are subject to changing volatility have fat tails and the question therefore arises as to what other distributions might be entertained and what is the behaviour of their score functions. One possibility is to assume a general error distribution (GED) leading to the Gamma-GED-EGARCH model. This model has a gamma distributed score and hence its properties may be obtained in much the same way as may the properties of Beta-t-EGARCH. However, although the GED family provides a compromise between the normal and t-distributions, the behaviour of its score function is not ideal from the point of view of robustness. We argue here that a better choice is a new model based on the family of exponential generalized beta distributions of the second kind (EGB2). The EGB2 distribution is light-tailed, but with higher kurtosis than the normal. It has featured in GARCH models before; see Wang et al (2001). But it has not been used in a DCS-EGARCH model and its score function has a form which makes it the ideal complement to Beta-t-EGARCH.

The first contribution of this paper is on what we shall refer to as the EGB2-EGARCH model. We derive its properties and contrast them with those of Betat-EGARCH and Gamma-GED-EGARCH. The asymptotic distribution of the maximum likelihood estimator of the dynamic parameters can be derived for EGB2-EGARCH just as it can for the other two models. An analytic expression for the asymptotic covariance matrix can be obtained and the conditions for the asymptotic theory to be valid are easily checked. The theory is much more straightforward than it is for the corresponding GARCH model.

The second aspect of the paper concerns tail indices in financial time series. Tail indices, which are a key feature of fat-tailed distributions, are often computed, and low values are cited as evidence of fat tails and the associated non-existence of higher moments¹. However, although excess kurtosis is a well-established stylized fact for both unconditional and conditional distributions of financial returns, the issue of whether such series have fat tails is more problematic. While it is undeniable that low tail estimates are a feature of financial returns, we argue that this does not, in itself, provide conclusive evidence of fat-tailed distributions. A subsidiary theme concerns the use of tail index estimates as starting values for the shape parameters of EGB2 and t-distributions, and in fact this provides a convenient lead-in to the discussion.

The article is organized as follows. Section 2 discusses classifications of tail

¹For example, Loretan and Phillips (1994) report (modified) Hill's estimates of between 3 and 4 for the unconditional distributions of many daily and monthly stock and exchange rate returns series. They conclude that fourth moments do not exist for such series.

behaviour in distributions and makes an important connection with the conditional score function. Section 3 describes the Beta-t-EGARCH and Gamma-GED-EGARCH models, and the associated asymptotic theory for the maximum likelihood estimator. The DCS scale model with an EGB2 distribution is analysed in Section 4 and the asymptotic theory is shown to extend to this case. Fitting Beta-t-EGARCH and EGB2-EGARCH to various returns series in Section 5 indicates that in a significant number of cases the EGB2 model gives a better fit, indicating that the conditional distribution does not have heavy tails. Since a DCS-EGARCH model cannot induce fat-tails, there is a paradox to be resolved. Section 6 analyses the problem and offers an explanation. Section 7 concludes.

2 Tails and tail indices

The Gaussian distribution has kurtosis of three and a distribution is said to exhibit *excess kurtosis* if its kurtosis is greater than three. Although many researchers take excess kurtosis as defining heavy tails, it is not, in itself, an ideal measure, particularly for asymmetric distributions. Most classifications in the insurance and finance literature begin with the behaviour of the upper tail for a non-negative variable, or one that is only defined above a minimum value; see Asmussen (2003) or Embrechts, Kluppelberg and Mikosch (1997). The two which are relevant here are as follows.

A distribution is said to be *heavy-tailed* if

$$\lim_{y \to \infty} \exp(y/\alpha) \overline{F}(y) = \infty \quad \text{for all } \alpha > 0, \tag{1}$$

where $\overline{F}(y) = \Pr(Y > y) = 1 - F(y)$ is the survival function. When y has an exponential distribution, $\overline{F}(y) = \exp(-y/\alpha)$, so $\exp(y/\alpha)\overline{F}(y) = 1$ for all y. Thus

the exponential distribution is not heavy-tailed.

A distribution is said to be *fat-tailed* if, for a fixed positive value of η ,

$$\overline{F}(y) = cL(y)y^{-\eta}, \qquad \eta > 0, \tag{2}$$

where c is a non-negative constant and L(y) is slowly varying, that is $\lim_{y\to\infty} (L(ky)/L(y)) = 1, k > 0.$

The parameter η is the (right) tail index. The implied PDF is a power law PDF

$$f(y) \sim cL(y)\eta y^{-\eta-1}, \qquad as \ y \to \infty, \quad \eta > 0,$$
 (3)

where \sim is defined such that $a(x) \sim b(x)$ as $x \to x_0$ if $\lim_{x\to x_0} (a/b) \to 1$. The m-th moment exists if $m < \eta$. The Pareto distribution is a simple case in which $\overline{F}(y) = y^{-\eta}$ for y > 1. If a distribution is fat-tailed then it must be heavy-tailed; see Embrechts, Kluppelberg and Mikosch (1997, p 41-2). On the other hand, not all heavy-tailed distributions are fat-tailed; the lognormal is an example.

The complement to the Pareto distribution is the power function distribution for which $\overline{F}(y) = y^{\overline{\eta}}$, 0 < y < 1, $\overline{\eta} > 0$. More generally,

$$\overline{F}(y) = cL(y)y^{\overline{\eta}}, \qquad 0 < y < 1, \quad \overline{\eta} > 0,$$

where $\overline{\eta}$ is the left tail index. Hence $f(y) \sim cL(y)\overline{\eta}y^{\overline{\eta}-1}$ as $y \to 0$.

The above criteria are related to the behavior of the conditional score and whether or not it discounts large observations. This, in turn, connects to robustness, as shown in Caivano and Harvey (2014). More specifically, consider a power law PDF, (3), with y divided by a scale parameter, φ , so that $\overline{F}(y/\varphi) = cL(y/\varphi)(y/\varphi)^{-\eta}$ and $f(y) \sim cL(y)\varphi^{-1}\eta(y/\varphi)^{-\eta-1}$. Then

$$\partial \ln f / \partial \varphi \sim \eta / \varphi \quad as \quad y \to \infty$$
(4)

and so the score is bounded. With the exponential link function, $\varphi = \exp(\lambda)$, $\partial \ln f / \partial \lambda \sim \eta$ as $y \to \infty$. Similarly as $y \to 0$, $\partial \ln f / \partial \lambda \sim \overline{\eta}$.

The logarithm of a variable with a fat-tailed distribution has exponential tails. Let x denote a variable with a fat-tailed distribution in which the scale is written as $\varphi = \exp(\mu)$ and let $y = \ln x$. Then for large y

$$f(y) \sim cL(e^y)\eta e^{-\eta(y-\mu)}, \qquad \eta > 0, \quad as \quad y \to \infty,$$

whereas as $y \to -\infty$, $f(y) \sim cL(e^y)\overline{\eta}e^{\overline{\eta}(y-\mu)}$, $\overline{\eta} > 0$. Thus y is not heavy-tailed but it may exhibit excess kurtosis. The score with respect to location, μ , is the same as the original score with respect to the logarithm of scale and so tends to η as $y \to \infty$. If a scale parameter is introduced, its score is bounded when divided by the variable.

The relevance of the above paragraph to this paper is that the (light-tailed) EGB2 variable is obtained by taking the logarithm of a fat-tailed GB2 variable.

3 Dynamic conditional score volatility models

A volatility model is typically of the form

$$y_t = \mu + \varphi_{t|t-1}\varepsilon_t, \qquad t = 1, ..., T, \tag{5}$$

where $\varphi_{t|t-1}$ is a time-varying scale and ε_t is a standardized IID random variable. The scale, $\varphi_{t|t-1}$, is proportional to $\sigma_{t|t-1}$, with the factor of proportionality depending on the shape parameter(s) of the distribution of ε_t . In a DCS model the dynamics are set up by letting the logarithm of a time-varying scale parameter be a linear function of the conditional score. In the case of first-order dynamics,

$$\lambda_{t+1|t} = \omega(1-\phi) + \phi \lambda_{t|t-1} + \kappa u_t, \tag{6}$$

where u_t is the score with respect to $\lambda_{t|t-1} = \ln \varphi_{t|t-1}$. Extensions to higher order models, components, seasonals and explanatory variables are discussed in Harvey (2013, ch 4).

The above model belongs to the EGARCH class introduced by Nelson (1991). The usual formulation has u_t replaced by $|\varepsilon_t|$. Moments of y_t exist for a GED distribution (with the normal being a special case), but Student's t is not viable because y_t has no moments for finite degrees of freedom. The dynamic scale model overcomes this difficulty because the score is a linear function of a variable with a beta distribution.

Like the GED, the EGB2 distribution offers a continuum of distributions between the normal and Laplace. However, unlike the GED (with shape parameter greater than one), the score with respect to the scale parameter of the EGB2 is bounded when divided by the variable². The associated dynamic scale model is described in Section 4. (The dynamic EGB2 location model is discussed in Caivano and Harvey, 2014).

The GARCH-t model is widely used in empirical finance. The GARCH-EGB2 has been studied by Wang et al (2001) but is far less common. In both models, $y_t = \mu + \sigma_{t|t-1}\epsilon_t$ and the variance is driven by squared observations, that is

 $\sigma_{t+1|t}^2 = \delta + \beta \sigma_{t|t-1}^2 + \alpha y_t^2, \quad \alpha,\beta \geq 0, \ \delta > 0,$

²This property features in the robustness literature; see Maronna *et al* (2006, p 34-8).

or, in notation similar to that in (6),

$$\sigma_{t+1|t}^2 = \delta + \phi \sigma_{t|t-1}^2 + \kappa \sigma_{t|t-1}^2 \epsilon_t^2,$$

where $\delta = \omega_{\sigma}(1 - \phi)$, where $\phi = \alpha + \beta$ and $\kappa = \alpha$.

For the DCS-EGARCH models with t and GED conditional distributions, all moments of the score exist and the existence of moments of y_t is not affected by the dynamics. The same is true of the EGB2. On the other hand, the existence of moments for GARCH models is affected by the volatility; see, for example, Mikosch and Starica (2000).

3.1 Maximum likelihood estimation

The ML estimates of the parameters, $\boldsymbol{\psi} = (\kappa, \phi, \omega)'$, in a DCS model can be obtained by maximizing the log-likelihood function with respect to the unknown parameters. The asymptotic distribution of the ML estimator in the first-order case is derived in Harvey (2013). Define

$$a = \phi + \kappa E\left(\frac{\partial u_t}{\partial \mu}\right)$$

$$b = \phi^2 + 2\phi \kappa E\left(\frac{\partial u_t}{\partial \mu}\right) + \kappa^2 E\left(\frac{\partial u_t}{\partial \mu}\right)^2 \ge 0$$

$$c = \kappa E\left(u_t \frac{\partial u_t}{\partial \mu}\right),$$
(7)

where unconditional and conditional expectations are the same. When scale and shape parameters are known and b < 1, the information matrix for a single observation is time-invariant and given by

$$\mathbf{I}(\boldsymbol{\psi}) = \sigma_u^2 \mathbf{D}(\boldsymbol{\psi}),\tag{8}$$

where σ_u^2 is the information quantity for a single observation and

$$\mathbf{D}(\boldsymbol{\psi}) = \mathbf{D} \begin{pmatrix} \widetilde{\kappa} \\ \widetilde{\phi} \\ \widetilde{\omega} \end{pmatrix} = \frac{1}{1-b} \begin{bmatrix} A & D & E \\ D & B & F \\ E & F & C \end{bmatrix},$$
(9)

with

$$A = \sigma_u^2, \qquad B = \frac{\kappa^2 \sigma_u^2 (1 + a\phi)}{(1 - \phi^2)(1 - a\phi)}, \qquad C = \frac{(1 - \phi)^2 (1 + a)}{1 - a},$$

$$D = \frac{a\kappa \sigma_u^2}{1 - a\phi}, \qquad E = c(1 - \phi)/(1 - a) \qquad and \qquad F = \frac{ac\kappa(1 - \phi)}{(1 - a)(1 - a\phi)}.$$

The ML estimator is asymptotically normal with covariance matrix given by the inverse of (8).

The above result can be easily extended to include the estimation of additional fixed parameters, such as the degrees of freedom in a t-distribution. Let $\boldsymbol{\theta}$ denote a vector of parameters such that $\boldsymbol{\theta} = (\theta_1, \theta_2')'$. Suppose that θ_2 consists of $n - 1 \ge 1$ fixed parameters, while θ_1 is time-varying and depends on a set of parameters, $\boldsymbol{\psi}$. When the terms in the information matrix of the static model that involve θ_1 , including cross-products, do not depend on θ_1 ,

$$\mathbf{I}\begin{pmatrix}\psi\\\theta_2\end{pmatrix} = \begin{bmatrix} E\left(\frac{\partial \ln f_t}{\partial \theta_1}\right)^2 \mathbf{D}(\psi) & \mathbf{d}E\left(\frac{\partial \ln f_t}{\partial \theta_1}\frac{\partial \ln f_t}{\partial \theta_2'}\right)\\ E\left(\frac{\partial \ln f_t}{\partial \theta_1}\frac{\partial \ln f_t}{\partial \theta_2}\right)\mathbf{d}' & E\left(\frac{\partial \ln f_t}{\partial \theta_2}\frac{\partial \ln f_t}{\partial \theta_2'}\right) \end{bmatrix},$$
(10)

where $\mathbf{D}(\boldsymbol{\psi})$ is the matrix in (9) and $\mathbf{d} = (0, 0, (1 - \phi)/(1 - a))'$. When the asymptotic distributions of the ML estimators of θ_1 and θ_2 are independent, the information matrix is block-diagonal and the top left hand block is as in (8).

3.2 Beta-t-EGARCH

The t_{ν} -distribution with a location of μ and scale of φ has probability density function

$$f(y;\mu,\varphi,\nu) = \frac{\Gamma\left(\left(\nu+1\right)/2\right)}{\Gamma\left(\nu/2\right)\varphi\sqrt{\pi\nu}} \left(1 + \frac{(y-\mu)^2}{\nu\varphi^2}\right)^{-(\nu+1)/2}, \qquad \varphi,\nu > 0,$$

where ν is the degrees of freedom and $\Gamma(.)$ is the gamma function. Moments exist only up to and including $\nu - 1$. For $\nu > 2$, the variance is $\sigma^2 = \{\nu/(\nu - 2)\}\varphi^2$. The excess kurtosis is $6/(\nu - 4)$, provided that $\nu > 4$. The t_{ν} distribution has fat tails for finite ν with the tail index given by ν ; see McNeil et al (2005, p 293).

When ε_t in (5) is t_{ν} distributed with $\mu = 0$ and $\varphi = 1$, the conditional score for the time-varying parameter $\lambda_{t,t-1}$ is

$$u_t = \frac{(\nu+1)y_t^2}{\nu \exp(2\lambda_{t:t-1}) + y_t^2} - 1 = \frac{(\nu+1)\varepsilon_t^2}{\nu + \varepsilon_t^2} - 1, \quad -1 \le u_t \le \nu, \quad \nu > 0.$$
(11)

At the true parameters values, u_t is IID and may be expressed as $u_t = (\nu + 1)b_t - 1$, where b_t is distributed as $beta(1/2, \nu/2)$; see Harvey (2013, ch 4). Analytic expressions for the moments and autocorrelations of y_t can be found from the infinite MA representation of $\lambda_{t|t-1}$. The asymptotic distribution for a stationary first-order model, as in (6), can be found from (8).

There are a number of ways in which skewness may be introduced into a tdistribution. One possibility is by the method proposed by Fernandez and Steel (1998); see Harvey and Sucarrat (2012).

3.3 Gamma-GED-EGARCH

The PDF of the general error distribution, denoted GED(v), is

$$f(y;\mu,\varphi,\upsilon) = \left[2^{1+1/\upsilon}\varphi\Gamma(1+1/\upsilon)\right]^{-1}\exp(-|(y-\mu)/\varphi|^{\upsilon}/2), \qquad \varphi,\upsilon > 0, \quad (12)$$

where φ is a scale parameter, related to the standard deviation by the formula $\sigma = 2^{1/v} (\Gamma(3/v) / \Gamma(1/v))^{1/2} \varphi$. The normal distribution is obtained when v = 2, in which case $\sigma = \varphi$. Setting v = 1 gives the Laplace, or double exponential, distribution, in which case $\sigma = 2\sqrt{2}\varphi$. Therefore when $1 \le v \le 2$ the GED distribution provides a continuum between the normal and Laplace. The kurtosis is $\Gamma(5/v) \Gamma(1/v) / \Gamma(3/v)$, so for v = 1 the excess kurtosis is three.

The conditional score for $\lambda_{t|t-1} = \ln \varphi_{t|t-1}$ is

$$u_t = (v/2) |y_t/\exp(\lambda_{t,t-1})|^v - 1, \qquad t = 1, ..., T.$$
(13)

The variable u_t is IID and may be expressed as $u_t = (v/2)g_t - 1$, where g_t has a gamma(2, 1/v) distribution. The score gives less weight to outliers than squared observations when v < 2, but it is not as robust as a Beta-t-EGARCH model with small degrees of freedom. Unlike the EGB2, the score is not bounded when divided by y_t , unless v = 1.

4 EGB2-EGARCH

The exponential generalized beta distribution of the second kind (EGB2) is obtained by taking the logarithm of a variable with a GB2 distribution. The distribution was first analyzed in Prentice (1975) and further explored by McDonald and Xu (1995). The PDF of a $GB2(\alpha, \nu, \xi, \varsigma)$ is³

$$f(x) = \frac{\nu(x/\alpha)^{\nu\xi-1}}{\alpha B(\xi,\varsigma) \left[(x/\alpha)^{\nu} + 1 \right]^{\xi+\varsigma}}, \qquad \alpha, \nu, \xi, \varsigma > 0,$$
(14)

where α is the scale parameter, ν, ξ and ς are shape parameters and $B(\xi, \varsigma)$ is the beta function. GB2 distributions are fat tailed for finite ξ and ς with upper and lower tail indices of $\eta = \varsigma \nu$ and $\overline{\eta} = \xi \nu$ respectively. The absolute value of a t_f variate is GB2($\varphi, 2, 1/2, f/2$) with tail index $\eta = \overline{\eta} = f$.

If x is distributed as $\text{GB2}(\alpha, \nu, \xi, \varsigma)$ and $y = \ln x$, the PDF of the $\text{EGB2}(\mu, \nu, \xi, \varsigma)$ variate y is

$$f(y;\mu,\nu,\xi,\varsigma) = \frac{\nu \exp\{\xi(y-\mu)\nu\}}{B(\xi,\varsigma)(1+\exp\{(y-\mu)\nu\})^{\xi+\varsigma}}.$$
(15)

The parameter which was the logarithm of scale in GB2 now becomes location in EGB, that is $\ln \alpha$ becomes μ . Furthermore ν is now a scale parameter, but ξ and ς are still shape parameters and they determine skewness and kurtosis.

4.1 Properties of EGB2

All moments of the EGB2 distribution exist. The first four are as follows:

Mean:
$$E(y) = \mu + \nu^{-1}[\psi(\xi) - \psi(\varsigma)]$$
 (16)

Variance:
$$\sigma^2 = E(y - E(y))^2 = \nu^{-2}[\psi'(\xi) + \psi'(\varsigma)]$$
 (17)

³The GB2 is described in Kleiber and Kotz (2003, ch6). Note that their convention has the order of α and ν reversed.

Skewness:
$$\frac{E(y - E(y))^3}{\sigma^3} = \frac{\psi''(\xi) - \psi''(\varsigma)}{[\psi'(\xi) + \psi'(\varsigma)]^{3/2}}$$
(18)

Kurtosis:
$$\frac{E(y - E(y))^4}{\sigma^4} = \frac{\psi'''(\xi) + \psi'''(\varsigma)}{[\psi'(\xi) + \psi'(\varsigma)]^2} + 3,$$
 (19)

where ψ , ψ' , ψ'' and ψ''' are polygamma functions of order 0, 1, 2 and 3 respectively. The EGB2 distribution is positively (negatively) skewed when $\xi > \zeta$ ($\xi < \zeta$) and its kurtosis decreases as ξ and ζ increase. Skewness ranges between -2 and 2 and kurtosis⁴ lies between 3 and 9. There is excess kurtosis for finite ξ and/or ζ .

Although ν is a scale parameter, it is the inverse of what would normally be considered a conventional measure of scale. Thus scale is better defined as $1/\nu$ or as the standard deviation

$$\sigma = \sqrt{\psi'(\xi) + \psi'(\varsigma)}/\nu = h(\xi,\varsigma)/\nu = h/\nu.$$
(20)

Thus

$$f(y;\mu,\sigma,\xi,\varsigma) = \frac{h\exp\{\xi h(y-\mu)/\sigma\}}{\sigma B(\xi,\varsigma)(1+\exp\{h(y-\mu)/\sigma\})^{\xi+\varsigma}}.$$

When $\xi = \varsigma$, the distribution is symmetric; for $\xi = \varsigma = 1$ it is a logistic distribution and when $\xi = \varsigma \to \infty$ it tends to a normal distribution. When $\xi = \varsigma = 0$ in the EGB2, the distribution is double exponential or Laplace; see Caivano and Harvey (2014). The following results will be used in a number of places when $\xi = \varsigma$: (i) $\xi h^2 = 2$ as $\xi \to \infty$, and $\xi h \to 2/h \to \infty$,(ii) $\xi h = \sqrt{2}$ for $\xi = 0$. Equivalently: (i) $\xi \psi'(\xi) = 1$ as $\xi \to \infty$, (ii) $\xi \sqrt{\psi'(\xi)} = 1$ for $\xi = 0$.

A plot of the (symmetric) EGB2, GED and Student's t with the same excess kurtosis shows them to be very similar. It is difficult to see the heavier tails of

⁴The maximum kurtosis in the symmetric case is 6 and is for $\xi = \varsigma = 0$. The kurtosis of 9 is achieved when ξ (or ς) = 0 and ς (or ξ)= ∞ .



Figure 1: PDF at the mean for GED, EGB2 and t-distributions with the same kurtosis.

the t distribution from the graph, and the only discernible difference among the three distributions is in the peak, which is higher and more pointed for the GED. The EGB2 in turn is more peaked than the t. As the excess kurtosis increases, the differences between the peaks become more marked; see Figure 1.

4.2 Dynamic scale model

The first-order dynamic scale model with EGB2 distributed errors is (5) where ε_t is a standardized ($\mu = 0, \nu = 1$) EGB2, that is $\varepsilon_t \sim EGB2(0, 1, \xi, \varsigma)$. Thus the conditional distribution is

$$f_t(y_t; \mu, \psi, \xi, \varsigma) = \frac{\exp\{\xi(y_t - \mu)e^{-\lambda_{t|t-1}}\}}{e^{\lambda_{t|t-1}}B(\xi, \varsigma)(1 + \exp\{(y - \mu)e^{-\lambda_{t|t-1}}\})^{\xi+\varsigma}},$$

where ψ now denotes the parameters in (6). The conditional score is

$$u_t = \frac{\partial \ln f(y_t)}{\partial \lambda_{t|t-1}} = (\xi + \varsigma)\varepsilon_t b_t - \xi\varepsilon_t - 1, \qquad (21)$$

where $\varepsilon_t = (y_t - \mu)e^{-\lambda_{t|t-1}}$ and

$$b_t = \frac{\exp\{(y-\mu)e^{-\lambda_{t|t-1}}\}}{1+\exp\{(y-\mu)e^{-\lambda_{t|t-1}}\}} = \frac{\exp\varepsilon_t}{1+\exp\varepsilon_t}.$$

At the true parameters values, $b_t \sim beta(\xi, \varsigma)$.

The model may be parameterized in terms of the standard deviation, $\sigma_{t|t-1}$, by defining $\epsilon_t = \varepsilon_t/h$. Then

$$y_t = \mu + \exp(\lambda_{\sigma,t|t-1})\epsilon_t, \qquad t = 1, ..., T,$$

with the only difference between $\lambda_{\sigma,t|t-1}$ and $\lambda_{t|t-1}$ being in the constant term which in $\lambda_{\sigma,t|t-1}$ is $\omega_{\sigma} = \omega + \ln h$; see the earlier discussion in sub-section 5.1. Note that the variance of ϵ_t is unity.

Writing the score, (21) as

$$u_t = h(\xi + \varsigma)\epsilon_t b_t - h\xi\epsilon_t - 1, \qquad (22)$$

it can be seen⁵ that when $\xi = \varsigma = 0, \sqrt{2} |\epsilon_t| - 1$ and, when $\xi = \varsigma \to \infty, u_t = \epsilon_t^2 - 1$.

Figure 2 compares the way observations are weighted by the score of an EGB2 distribution with $\xi = \varsigma = 0.5$, a Student's t_7 distribution and a GED(1.148). These are the same distributions used in Figure 1; all have excess kurtosis of 2. Dividing (22) by ϵ_t gives a bounded function as $|\epsilon_t| \to \infty$. This is consistent with the 'soft'

⁵When $\xi = 0$, $\xi h = \sqrt{2}$ and b_t degenerates to a Bernoulli variable such that $b_t = 0$ when $\epsilon_t < 0$ and $b_t = 1$ when $\epsilon_t > 0$. Then $2b_t - 1 = 1$ (-1) for $\epsilon_t > 0$ ($\epsilon_t < 0$) and the score can be written as: $u_t = \sqrt{2} |\epsilon_t| - 1$.

As regards $\xi \to \infty$, note that because $\partial b_t / \partial \epsilon_t = hb_t (1 - b_t)$, a first order Taylor expansion of b_t around $\varepsilon_t = 0$ yields $b_t \simeq \frac{1}{2} + \frac{h}{4} \epsilon_t$. Therefore $2b_t - 1 \simeq (h/2)\epsilon_t$ and $u_t \simeq (\xi h^2/2)\epsilon_t^2 - 1$. As $\xi \to \infty$, $\xi h^2 \to 2$.



Figure 2: Score functions for EGB2 (thick line), GED (medium line) and t (thick dash), all with unit variance and an excess kurtosis of 2. Thin dash shows normal score. (These score functions are even).

Winsorizing⁶ of the location score; see Caivano and Harvey (2014).

The unconditional mean is given by $E(y_t) = \mu + E(\varepsilon_t) E(e^{\lambda_{t|t-1}})$, whereas the m-th unconditional moment about the mean is $E(\varepsilon_t^m) E(e^{m\lambda_{t|t-1}})$, m > 1. In the Betat-EGARCH and Gamma-GED-EGARCH models analysed in Harvey (2013, ch4), the expression $E(\exp(m\lambda_{t|t-1}))$ depends on the moment generating functions (MGF) of beta and gamma variates, respectively, which have a known form. For EGB2-EGARCH, the unconditional moments depend on the MGF of u_t , ie $E_{EGB2(\xi,\varsigma)}[mu_t]$, where u_t is defined in (21). For the limiting normal and Laplace cases of the EGB2, the score functions and hence the unconditional moments are the same as for v = 2and v = 1 in Gamma-GED-EGARCH; see Harvey (2013, sub-section 4.2.2). For v = 1 it is necessary to have $m\kappa < 1$ in the first-order model for the m-th moment

⁶The M-estimator, which features prominently in the robustness literature, has a Gaussian response until a certain threshold, K, whereupon it is constant; see Maronna *et al* (2006, p 25-31). This is known as Winsorizing as opposed to trimming where observations greater in absolute value than K are set to zero.

to exist, whereas for v = 2 the condition is $m\kappa < 1/2$. For $0 < \xi, \varsigma < \infty$ having the last condition hold is therefore sufficient for the existence of the unconditional moments. This being the case, we can at least assert, from Jensen's inequality, that the unconditional moments exceed the conditional moments and that the kurtosis increases; see Harvey (2013, p 102).

The MGF of u_t is also required to find the conditional expectations needed to forecast volatility and volatility of volatility. However, it is the full ℓ – *step* ahead conditional distribution which is often needed in practice and this is easily simulated from standardized beta variates. The quantiles, such as those needed for VaR and the associated expected shortfalls, may be estimated at the same time.

4.3 Maximum likelihood estimation

The asymptotic distribution of the ML estimators of the parameters in a dynamic scale model with a symmetric EGB2 distribution is given in the proposition below. The score and its derivatives are linear combinations of variables of the form $\varepsilon_t^r b_t^h (1 - b_t)^k$, r, h, k = 0, 1, 2.. and the properties of these variables are such that the conditions for convergence and asymptotic normality of the maximum likelihood estimator may be verified without too much difficulty. The formulae for the general result on the asymptotic distribution are quite complex; see and Harvey (2014).

Proposition 1 Suppose that ε_t in (5) is known to be symmetric with a standardized $EGB2(0, 1, \xi, \xi)$ distribution. Let $\lambda_{t|t-1}$ be generated by (6) with $|\phi| < 1$. Define a, b and c as in (7) with

$$E(u'_t) = \frac{1 - 2\xi^2 \psi'(\xi) - 2\xi}{2\xi + 1} = -\sigma_u^2$$
(23)

$$E(u_t^{\prime 2}) = \frac{\xi^3 \left(\xi + 1\right)}{\left(2\xi + 3\right) \left(2\xi + 1\right)} \left(2\psi^{\prime\prime\prime}(\xi + 2) + 12\psi^{\prime 2}(\xi + 2)\right) + \sigma_u^2 + 1 \tag{24}$$

and

$$E(u_t u_t') = -1. \tag{25}$$

Let $\boldsymbol{\psi} = (\kappa, \phi, \omega)'$. Assuming that b < 1 and $\kappa \neq 0$, $(\mu, \widetilde{\boldsymbol{\psi}}', \widetilde{\boldsymbol{\xi}},)'$, the ML estimator of $(\boldsymbol{\mu}, \boldsymbol{\psi}', \boldsymbol{\xi})'$, is consistent and the limiting distribution of $\sqrt{T}(\widetilde{\mu} - \mu, (\widetilde{\boldsymbol{\psi}} - \boldsymbol{\psi})', \widetilde{\boldsymbol{\xi}} - \boldsymbol{\xi})'$ is multivariate normal with mean vector zero and covariance matrix given by $Var(\widetilde{\mu}, \widetilde{\boldsymbol{\psi}}, \widetilde{\boldsymbol{\xi}},) = \mathbf{I}^{-1}(\mu, \boldsymbol{\psi}, \boldsymbol{\xi})$, where the information matrix is

$$\mathbf{I}\begin{pmatrix} \mu\\ \psi\\ \xi \end{pmatrix} = \begin{bmatrix} \frac{\xi^2}{1+2\xi} E(e^{-2\lambda_{t|t-1}}) & 0 & 0\\ 0 & \frac{2\xi+2\xi^2\psi'(\xi)-1}{1+2\xi} \mathbf{D}(\psi) & -\frac{1}{\xi}\mathbf{d}\\ 0 & -\frac{1}{\xi}\mathbf{d}' & 2\psi'(\xi) - 4\psi'(2\xi) \end{bmatrix}.$$
 (26)

The block diagonality of (26) means that the asymptotic variances of ξ and the parameters in ψ can be computed even though an expression for the unconditional expectation of $\exp(\lambda_{t|t-1})$ is difficult to derive for $0 < \xi < \infty$.

Remark 1 The information matrix is more complicated if ω_{σ} (which is $\omega + \ln h$) is used rather than ω (although it can still be found). However, standard errors are of little practical importance for the constant term and the standard errors of the other parameters do not depend on its parameterization.

When $\xi = 0$, so that the distribution is Laplace, $E(u'_t) = -1$. Similarly as $\xi \to \infty$, $E(u'_t) = -2$, which is the correct result for a Gaussian distribution. In addition, when $\xi = 0$, both $\psi'(\xi + 2)$ and $\psi'''(\xi + 2)$ are finite, so $E(u'_t) = 2$. Hence

$$b = \phi^2 - 2\phi\kappa + 2\kappa^2, \tag{27}$$

which is the same as given by the expression in Harvey (2013, p 120) for b in Gamma-GED-EGARCH when v = 1. (Also c = -1.) Similarly for $\xi \to \infty$,

$$b = \phi^2 - 4\phi\kappa + 12\kappa^2$$

5 Exchange rates

Tables 1 and 2 report the full ML estimates of the (symmetric) EGB2-EGARCH and Beta-t-EGARCH models for the returns of exchange rates of developed and emerging countries against the US dollar. Developed countries currencies include the Australian dollar (AUD), the Canadian dollar (CAD), the Swiss franc (CHF), the Denmark krone (DKK), the Euro (EUR), the Pound sterling (GBP), the Japanese yen (JPY), the Norwegian krone (NOK), the New Zealand dollar (NZD) and the Swedish krona (SEK). Emerging countries currencies include the Brazilian Real (BRL), the Chinese renmimbi (CNY), the Hong Kong dollar (HKD), the Indian rupee (INR), the South Korean won (KRW), the Sri Lanka rupee (LKR), the Mexican peso (MXN), the Malaysian ringgit (MYR), the Singapore dollar (SGD) the Thai baht (THB), the Taiwan dollar (TWD) and the South African Rand (ZAR). Exchange rate data are daily and range from 4th January 1999 to 15th March 2013.

As can be seen, the EGB2 gives a better fit for five developed countries, whereas the t is best for four. In the case of Switzerland, the exchange rate experienced a sudden fall on 6th September 2011 when the Swiss National Bank announced its intention to enforce a ceiling on the exchange rate of the euro against the Swiss franc. If the resulting outlier is removed from the returns series, the EGB2 performs better than the Student's t. For the developing countries the situation is very different in that the EGB2 is better than the t in only three cases out of 12. For four currencies

the	estimated	degrees	of freedo	om of	f the	t-distrib	ution	is	below	three	and	in	these
cas∈	s the ML	estimatio	on of the	EGI	32 m	odel faile	ed to d	con	verge ⁷				

			EGB	2		t					
	κ	ϕ	ω	ξ	Log-L	κ	ϕ	ω	ν	Log-L	
AUD	0.030	0.991	-5.34	1.29	12523.7	0.030	0.992	-5.04	9.22	12526.2	
CAD	0.023	0.996	-5.55	1.71	13823.2	0.024	0.996	-5.40	12.64	13822.4	
CHF	0.018	0.993	-5.56	1.05	12848.2	0.017	0.994	-5.14	8.47	12849.9	
CHF^*	0.017	0.994	-5.47	1.22	12865.0	0.016	0.994	-5.13	9.69	12863.1	
DKK	0.019	0.995	-5.61	1.12	13086.9	0.018	0.995	-5.22	8.87	13086.9	
EUR	0.017	0.995	-5.43	1.46	13118.4	0.017	0.995	-5.20	11.13	13117.4	
GBP	0.022	0.994	-5.30	2.30	13575.8	0.022	0.994	-5.33	16.01	13575.3	
JPY	0.024	0.989	-5.87	0.73	13074.7	0.024	0.990	-5.21	6.14	13078.2	
NOK	0.018	0.997	-5.38	1.54	12596.8	0.018	0.997	-5.19	11.03	12596.1	
NZD	0.024	0.992	-5.41	1.03	12184.0	0.024	0.992	-4.98	7.77	12184.7	
SEK	0.018	0.996	-5.15	1.91	12547.4	0.018	0.996	-5.07	13.18	12546.9	

* CHF series without the outlier corresponding to September 6, 2011

Table 1 ML estimates for exchange rate data (developed countries)

⁷Although it is not the purpose of this exercise to compare DCS EGARCH models with standard GARCH - there is already a good deal of evidence in Creal et al (2011), Harvey and Sucarrat (2012) and elsewhere to suggest that DCS EGARCH tends to be better- we did fit GARCH-t models and found that in only 7 out of 23 cases did they beat Beta-t-EGARCH.

			EGB2	2		Т					
	κ	ϕ	ω	ξ	Log-L	κ	ϕ	ω	ν	Log-L	
BRL	0.082	0.975	-5.25	1.08	12099.7	0.087	0.974	-484	8.30	12098.8	
CNY	0.011	0.998	-9.73	0.35	23414.1	0.057	0.999	-9.38	5.37	23919.2	
HKD	-	-	-	-	-	0.213	0.985	-9.15	2.29	26527.7	
INR	-	-	-	-	-	0.102	0.992	-6.49	2.97	16136.4	
MYR	0.024	1.000	-10.11	1.16	21792.3	0.078	1.000	-9.81	9.64	21907.8	
MXN	0.055	0.979	-5.63	1.31	13727.8	0.055	0.980	-5.34	9.32	13728.9	
ZAR	0.042	0.991	-5.09	1.18	11596.4	0.042	0.991	-4.74	8.75	11596.5	
SGD	0.032	0.989	-6.48	0.89	15724.7	0.033	0.989	-5.95	7.04	15726.9	
KRW	0.074	0.985	-6.75	0.33	14026.1	0.074	0.984	-5.49	4.83	14020.1	
LKR	-	-	-	-	-	0.167	0.974	-7.08	1.89	18451.6	
TWD	-	-	-	-	-	0.111	0.974	-6.47	2.72	16727.3	
THB	0.096	0.968	-7.25	0.31	15410.3	0.091	0.970	-5.95	4.26	15405.3	

Table 2 ML estimates for exchange rate data (emerging countries)

6 Scale parameters and tail indices

Tail index estimators may be computed prior to fitting DCS-EGARCH volatility models. As such they may be used as starting values for an iterative maximum likelihood estimation procedure. Sub-section 6.1 reviews tail estimators and subsection 6.2 presents evidence on the accuracy with which they may be expected to estimate the scale parameters of an EGB2 distribution when applied to the residuals from fitting a preliminary model to returns. A similar analysis is conducted on the estimation of the degrees of freedom of a t-distribution from tail indices computed from the logarithms of absolute returns. Sub-section 6.3 returns to the exchange rate data of Section 5 and presents estimates of the tail indices, and implied shape parameters, computed from the residuals from GARCH models. These estimates are much smaller than the corresponding ML estimates.

The use of residuals from a preliminary model can be avoided simply by using the raw data on returns because, in theory, the tail index estimators will still be consistent; see Resnick and Starica (1995). However, it seems that the increased kurtosis induced by dynamic volatility can substantially increase the downward bias. These findings have important implications for the conclusions to be drawn from estimating tail indices by nonparametric methods.

6.1 Tail index estimators

Hill's estimator of the tail index for a fat-tailed distribution is

$$\widehat{\eta} = \left(k^{-1}\sum_{j=1}^{k}\ln x_j - \ln x_k\right)^{-1} = \left(k^{-1}\sum_{j=1}^{k}y_j - y_k\right)^{-1}$$

where x_j and y_j , j = 1, ..., k, denote the observations in descending order. Embrechts, Kluppelberg and Mikosch (1997, p 336-7) set out the asymptotic properties for a power law distribution of the form (3). The variance of the limiting (normal) distribution of $\sqrt{k}(\hat{\eta} - \eta)$ is η^2 , so the asymptotic variance of $\ln \hat{\eta}$ is 1/k. Note that the asymptotic theory requires not only that T and $k \to \infty$, but that $k/T \to 0$.

A similar estimator, $\hat{\eta}$, may be constructed for the lower tail index by putting the observations in ascending order and using the smallest observations. When the observations come from a (symmetric) distribution, an estimate of location is subtracted and Hill's estimator is then constructed from the logarithms of absolute values. It is well-known that Hill's estimator can be quite badly biased; it is usually too low. Various alternatives have been suggested, one of the more recent ones being the OLS estimator from a regression of log rank minus half on log size; see Gabaix and Ibragimov (2011). However, even the improved estimators have bias and this bias turns out to play an important role in influencing the conclusions that one might be tempted to draw.

Because the performance of both Hill's and OLS estimates improves the more observations are excluded from the tail, one might be tempted to exclude as many observations as possible. However, doing so can lead to very imprecise estimates. A careful choice of the truncation point is needed in order to achieve a good biasvariance trade-off; see the plots in Embrechts et al (1997).

6.2 Tail index estimators of shape parameters for EGB2 and Student t distributions

The upper and lower tail indices in the GB2 distribution are $\nu\varsigma$ and $\nu\xi$ respectively. Hence estimators of ς and ξ in the EGB2 model may be obtained from standardized residuals from an initial model by solving the equations $\hat{\eta} = h(\hat{\xi}, \hat{\varsigma})\hat{\varsigma}$ and $\hat{\overline{\eta}} = h(\hat{\xi}, \hat{\varsigma})\hat{\xi}$. Note that the lower bound for η (= $\overline{\eta}$) is obtained in the symmetric model when $\xi = \varsigma = 0$ and is $\sqrt{2}$. More generally the lower bound is one for $\hat{\eta}$ ($\hat{\overline{\eta}}$) when $\varsigma(\xi) = 0$ and $\xi(\varsigma) > 0$. There is no finite upper bound. In the symmetric case the tail index values implied by various values of $\xi = \varsigma$ – given in brackets - are as follows: 14.18 (100), 3.33 (5), 2.27 (2), 1.81 (1), 1.57 (0.5).

When Hill's estimator is constructed from the *logarithms* of absolute values of residuals, it gives an estimator of the degrees of freedom of a t-distribution directly.

In order to assess the accuracy of the Hill's and OLS estimators for the EGB2 and



Figure 3: Average estimates of tail index plotted against true tail index for a GB2 corresponding to $EGB2(0, 1, \xi, \xi)$.

t-distributions, simulations for T = 10,000 were carried out using 1,000 replications. The results for an $EGB2(0, 1, \xi, \xi)$ are shown in Figure 3; setting with $\nu = 1$ means that $\eta = \xi$. The OLS estimator dominates Hill's estimator in terms of bias, but it still underestimates the true tail index, with the bias increasing with the shape parameter. The bias also depends on how many observation are included in the tail: when 10% of the observations are included, the bias is already non-negligible for $\xi = 1.5$. On the other hand, if we include only 1% of the observations, the estimate is still relatively reliable when $\xi = 2$.

The ML estimates of the EGB2 shape parameters reported in Table 1 are all quite low. Hence the tail index estimates obtained by the Hill and OLS methods will provide good starting values for parameters of this order of magnitude. On the other hand, for a t distribution, the bias in Hill's estimator is large even for a relatively small degrees of freedom and a 1% truncation; see Figure 4. The bias becomes considerably worse as the degrees of freedom increase. The OLS estimator offers



Figure 4: Average estimates of tail index plotted against degrees of freedom for t_{ν} .

some improvement but not a great deal. The same is true of other modifications. Studies of new estimators are often confined to small tail indices; for example, Huisman at al (2001) only⁸ present results for a t-distribution with $\nu \leq 5$.

The differences in the value of the tail index estimators as starting values for t and EGB2 stems from the fact that the low values of the EGB2 shape parameter, ξ , correspond to tail indices for the GB2 distribution that are much smaller than the the tail indices for the t-distribution. Figure 5 shows the tail index estimators for EGB2 and t-distributions plotted in such a way that the values of ξ on the horizontal axis for EGB2 correspond to a tail index for GB2 that is similar to the degrees of freedom (and hence tail index) for the t.

The above graphs prompt the question as to the behaviour of tail estimators when the distribution does not have fat tails. An analysis of the log-normal distribution provides some insight. The log-normal distribution is sub-exponential⁹, but it is not fat-tailed and all its moments exist. Hence the tail index should theoretically be infinite. However, consider Hill's estimator which, as McNeil et al (2005, p 286-7)

⁸Nevertheless they conclude on p 214 that '..tail fatness is easily exaggerated in small samples.' ⁹See Embrechts, Kluppelberg and Mikosch (1997, p 34)



Figure 5: Hill's and OLS estimates compared for GB2 and t distributions.

observe, is motivated by the mean excess function of the logarithm of variable, x, with a fat-tailed distribution, that is

$$e(y^*) = E(y - y^* \mid y > y^*), \tag{28}$$

where $y = \ln x$. Hill's estimator is the inverse of the sample mean excess function. For a Pareto distribution, y is exponentially distributed and $e(0) = 1/\eta$ is just the mean. For the log-normal, we can make use of the relationship between $ES(\alpha)$, the expected shortfall for a Gaussian variable, that is $y \sim N(\mu, \sigma^2)$, beyond the α quantile, and the mean excess function. Specifically,

$$e(y^* = y_\alpha) = ES(\alpha) - \mu - z_\alpha \sigma,$$

where $y_{\alpha} = \mu + z_{\alpha}\sigma$ and z_{α} is the α quantile for a standard normal variate. From

the formula for $ES(\alpha)$ derived in McNeil et al (2005, p 45),

$$e(y_{\alpha}) = \sigma\left(\frac{\phi(z_{\alpha})}{1-\alpha} - z_{\alpha}\right),$$

where $\phi(z)$ is the PDF at z of a standard normal variate. Evaluating $1/e(y_{\alpha})$ then gives the p lim of Hill's estimator, which we will denote as H_{α} . Table 3 shows H_{α} multiplied by σ for typical values of α . The estimator improves, in the sense that its p lim gets bigger, as α gets smaller, which is consistent with $k/T \to 0$. As $\sigma \to 0$, the log-normal tends towards a (degenerate) normal and $H_{\alpha} \to \infty$. On the other hand, as σ increases, $H_{\alpha} \to 0$. The fall in H_{α} corresponds to an increase in excess kurtosis, which is $\exp(4\sigma^2) + 2\exp(3\sigma^2) + 3\exp(2\sigma^2) - 6$. For $\sigma = 0.5$, the excess kurtosis is 5.90 whereas the skewness, $(\exp(\sigma^2)+2)\sqrt{\exp(\sigma^2)-1}$, is 1.75. For $\sigma = 1$, the skewness is far more pronounced and the excess kurtosis is 110.94. Even with $\sigma = 0.5$, one might conclude, quite erroneously, that, on the basis of the 5% quantile, the existence of fifth moment, and perhaps even the fourth, is in doubt. Even setting α to the unrealistically small value of 0.001 gives a p lim of only twelve for Hill's estimator.

α	0.10	0.05	0.01	0.001
σH_{α}	2.10	2.39	3.23	6.00

Table 3 Plim of Hill's estimator (times σ) for data from a lognormal distribution for different quantiles, α .

The above analysis suggest that tail index estimates may be low even when the true index is infinite, with the index estimates being closer to zero the higher is the kurtosis. The average tail indices computed from the logarithms of absolute values of returns of a simulated EGB2 distribution are shown in Figure 6 and the results confirm this conjecture. For example, when $\xi = 1$, the Hill's estimates are centred on



Figure 6: Average estimates of tail index from the logarithms of absolute returns plotted against ξ when data is $EGB2(0, 1, \xi, \xi)$.

 $H_{0.05} \simeq 4.5$ while the corresponding figure for the OLS estimator is approximately 5.3.

6.3 Estimates of shape parameters from tail indices of residuals

Tables 4 and 5 compare the ML estimates of the shape parameters for EGB2-EGARCH and Beta-t-EGARCH models obtained in Section 5 with those implied by the Hill's and OLS estimates¹⁰ obtained from the standardized residuals of a GARCH(1,1) model (estimated by QML, assuming normality). For Beta-t-EGARCH the tail estimates are computed from the logarithms of absolute values. The implied EGB2 shape parameter is given by solving the equation $\eta = \xi \sqrt{2\psi'(\xi)}$, whereas the degrees of freedom for the t is the tail index. As might be expected from Figure 4,

¹⁰In order to choose the optimal truncation point for the Hill's and OLS estimators a commonly suggested strategy is to plot the estimators for various truncation points and to choose one in a region were the estimator is reasonably stable. A look at Hill's plots showed them to be very unstable in many cases. Nevertheless we report the maximum value obtained in this way (for thresholds less than 20%).

both Hill's and OLS estimates tend to be much smaller than the ML estimates of ν in the t-distribution. This is not true of the estimates for the EGB2 distribution, for the reasons given earlier. However, for emerging countries there are a number of missing entries for EGB2 because the corresponding tail index estimate was below the theoretical lower bound of $\sqrt{2}$; it comes as no surprise that most of these occur when the ML procedure failed to converge. In such cases there is a clear indication of fat tails.

	Hill's				OLS		EGB2-EGARCH		
	5%	10%	max	5%	10%	max	ξ	Implied kurtosis	
AUD	0.93	1.08	1.11	-	0.20	0.56	1.29	0.94	
CAD	1.73	1.60	1.93	1.16	1.41	1.45	1.71	0.70	
CHF	1.24	1.36	1.40	-	0.57	0.82	1.05	1.15	
DKK	1.46	1.12	1.47	0.46	0.80	0.95	1.12	1.08	
EUR	2.10	1.44	2.10	1.50	1.58	1.60	1.46	0.83	
GBP	2.03	1.55	2.11	2.00	1.85	2.35	2.30	0.51	
JPY	0.35	0.53	0.73	-	-	0.28	0.73	1.56	
NOK	1.19	1.15	3.28	1.79	1.34	2.88	1.54	0.78	
NZD	0.56	0.86	0.98	0.17	0.44	0.65	1.03	1.17	
SEK	1.53	1.23	2.57	2.20	1.61	3.29	1.91	0.62	

Developed countries

Table 4a Shape parameter estimates implied by tail index estimates for EGB2-

EGARCH.

		Hill's			OLS		Beta-t-EGARCH				
	5%	10%	max	5%	10%	\max	ν	Implied kurtosis			
AUD	4.52	3.88	4.70	4.30	4.23	4.33	9.22	1.15			
CAD	5.20	4.30	6.60	5.54	5.07	6.55	12.64	0.69			
CHF	4.79	4.17	5.90	4.98	4.68	5.01	8.47	1.34			
DKK	5.11	3.88	6.19	5.19	4.72	5.20	8.87	1.23			
EUR	5.68	4.16	6.47	5.81	5.14	6.73	11.13	0.84			
GBP	5.49	4.24	7.17	6.13	5.24	8.05	16.01	0.50			
JPY	3.89	3.33	4.26	4.01	3.72	4.33	6.14	2.80			
NOK	4.64	3.92	8.99	5.92	4.65	9.10	11.03	0.85			
NZD	4.01	3.67	4.67	4.39	4.06	4.95	7.77	1.59			
SEK	4.98	3.95	7.52	6.33	4.85	9.32	13.18	0.65			

Developed countries

Table 4b Tail index (degrees of freedom) estimates for Student's t

		Hill's			OLS		EGB2-EGARCH						
	5%	10%	max	5%	10%	\max	ξ	Implied kurtosis					
BRL	0.68	0.85	1.13	0.12	0.45	0.70	1.08	1.12					
CNY	-	-	-	-	-	-	0.35	2.35					
HKD	-	-	-	-	-	-	-	-					
INR	-	-	-	-	-	-	-	-					
KRW	0.22	0.35	0.50	-	-	0.03	0.33	1.04					
LKR	-	-	-	-	-	-	-	0.93					
MXN	0.47	0.63	0.85	-	0.28	0.50	1.31	1.03					
MYR	-	-	-	-	-	-	1.16	1.33					
SGD	0.53	0.70	0.73	-	0.15	0.42	0.89	2.40					
THB	-	0.10	0.34	-	-	-	0.31	-					
TWD	-	-	-	-	-	-	-	-					
ZAR	0.84	0.99	1.27	0.92	0.89	1.25	1.18	2.45					

Emerging countries

Table 5a Shape parameter estimates implied by tail index estimates for EGB2-

EGARCH.

	0.0												
		Hill's			OLS		Beta-t-EGARCH						
	5%	10%	\max	5%	10%	\max	ν	Implied kurtosis					
BRL	4.00	3.44	4.83	4.11	3.89	5.37	8.30	1.40					
CNY	2.60	2.02	2.92	2.58	2.37	2.58	5.37	4.38					
HKD	2.42	2.14	2.85	2.39	2.33	2.39	2.29	-					
INR	2.98	2.54	3.24	3.09	2.85	3.18	2.97	-					
KRW	3.86	3.23	4.79	4.20	3.76	4.48	4.83	1.06					
LKR	2.53	2.24	2.65	2.39	2.42	2.45	1.89	1.13					
MXN	4.05	3.50	5.27	4.40	4.02	5.02	9.32	1.26					
MYR	3.22	2.53	3.97	3.30	3.05	3.32	9.64	1.97					
SGD	4.07	3.52	5.01	4.33	3.97	4.63	7.04	7.23					
THB	3.47	3.08	4.25	4.00	3.51	4.60	4.26	-					
TWD	3.05	2.65	3.33	3.12	2.93	3.18	2.72	-					
ZAR	4.25	3.74	6.18	5.00	4.30	6.91	8.75	23.08					

Emerging countries

Table 5b Tail index (degrees of freedom) estimates for Student's t

6.4 Tail index estimates for raw data

Although the tail index estimators are consistent when computed from raw data, they are typically much lower the corresponding estimates obtained from residuals. This is certainly true of the tail index estimates of the exchange rates of Section 5 as reported by Ibragimov et al. (2013). Even for the developed economies, the tail index estimates are mostly less than four, implying infinite fourth moments; see also Loretan and Phillips (1994).

There is some work to suggest that for fat-tailed conditional distributions, a

GARCH(1,1) process can lower the tail index when it is close to IGARCH; see Mikosch and Starica (2000) and Huisman et al (2001, p212). However, some calculations in McNeil et al (2005, p 296-7) suggest that, for plausible values of the parameters, the reduction may be small. For a stationary Beta-t-EGARCH the situation is perhaps more clear-cut in that a basic property of the model is that the existence of moments, and hence the tail index of the conditional distribution, is not changed by changing volatility. What does change, for EGB2 EGARCH as well as Beta-t-EGARCH, is that the excess kurtosis increases. The increase can be worked out and Table 6 shows the (proportional) increase for normal, Laplace and t-distributions. Perhaps surprisingly the increase is bigger for Laplace, and to a lesser extent normal, than it is for t when $\phi = 0.999$. It was noted in the previous sub-section that tail index estimates can be quite low even when EGB2 fits better than t, and so the fact that the increase in kurtosis can be very large for a Laplace distribution with persistent volatility is of some significance.

	Kurtosis	Incre	Increase in kurtosis, K							
κ	-	.03			.06					
ϕ	-	.98	.99	.999	.98	.99	.999			
normal	3	1.05	1.10	2.35	1.24	1.54	43.38			
Laplace	6	1.10	1.28	5.51	1.54	2.36	1881			
\mathbf{t}	6	1.25	1.34	1.64	1.74	2.69	8.52			

Table 6 Increase in kurtosis induced by changing volatility

7 Conclusions and extensions

Most financial returns time series exhibit non-normal behavior, which is often modeled by a Student t distribution. This choice is strongly supported by tail index estimates, which almost invariably point to fat-tailed distributions. We argue here that the fat-tailed distributions are not always appropriate and that for many returns series a leptokurtic distribution which is light-tailed can give a better fit. The EGB2 distribution provides a bridge between the normal and t-distribution in that it exhibits excess kurtosis without having heavy tails. Unlike the general error distribution (apart from Laplace), it has a score function that is bounded when divided by the variable. This property corresponds to the gentle form of Winsorizing that is a feature of the EGB2 score for location. Both EGB2 and a modified version of the t-distribution are able to handle asymmetric distributions.

The EGB2 and Beta-t-EGARCH models were fitted to data on exchange rates and stock returns. For the exchange rates of developed countries, the evidence for fat-tails is unconvincing. On the whole the EGB2 fits better than the t, with the tail indices computed for both residuals and raw data being entirely consistent with the kind of values indicated by our simulations. The case for fat-tails in the distributions of developing country exchange rates is more persuasive. Similarly for most stock prices a t-distribution seems to fit better than EGB2.

The raw tail indices are very misleading when the conditional distribution is not fat-tailed. Even when the conditional distribution is best modeled by a Student's t, tail index estimates are typically much smaller than the degrees of freedom estimated by maximum likelihood, probably because of the increase in kurtosis which changing volatility induces. The low tail indices should be treated with caution if conclusions about the existence of moments are to be drawn. On a more positive note, they can be useful as an indicator of fat-tailed distributions with very small tail indices. Similarly they can provide sensible starting values for shape parameters in the EGB2 distribution, because these parameters are typically quite small.

In summary, while it is undeniable that low tail index estimates are a feature of

financial returns, we argue that this does not, in itself, provide strong evidence of fat-tailed distributions. Our findings lend support to the cautionary note sounded by Clauset et al (2009) on this matter. Placing too much store on nonparametric estimates, particulary from raw data, is unwise.

Acknowledgement

We are grateful to Paul Kattuman for providing us with the exchange rate data.

REFERENCES

Asmussen, S. (2003). Applied Probability and Queues, Berlin, Springer Verlag.

- Caivano, M and A.C. Harvey (2014). Time series models with an EGB2 conditional distribution. Temi di discussione (Working papers) no 947. Bank of Italy. Submitted.
- Clauset, A., Shalizi, C. R. and M. E. J. Newman (2009). Power-Law Distributions in Empirical Data. SIAM Review, 51, 661–703.
- Creal, D., Koopman, S.J. and A. Lucas (2011). A Dynamic Multivariate Heavy-Tailed Model for Time-Varying Volatilities and Correlations, Journal of Business and Economic Statistics, 29, 552-63.
- Embrechts, P., Kluppelberg, C. and T. Mikosch (1997). Modelling Extremal Events. Berlin: Springer Verlag.
- Fernandez, C. and M. F. J. Steel (1998). On Bayesian modeling of fat tails and skewness. Journal of the American Statistical Association 99. 359-371.
- Gabaix, X. and R. Ibragimov (2011). Rank-1/2: a simple way to improve the OLS estimation of tail exponents. Journal of Business and Economic Statistics, 29, 24-39.
- Harvey, A. C. (2013). Dynamic models for Volatility and Heavy Tails. Econometric Society Monograph. Cambridge University Press.
- Harvey, A. C. and G. Sucarrat (2012). EGARCH Models with Fat Tails, Skewness and Leverage. Cambridge Working paper in Economics, CWPE 1236. Submitted.

- Huisman, R., Koedijk, K.G. Kool, C.J.M. and F. Palm (2001). Tail-index estimates in small samples. *Journal of Business and Economic Statistics*, 19, 208-16.
- Ibragimov, M., Ibragimov, R. and P. Kattuman. (2013). Emerging markets and heavy tails, Journal of Banking & Finance, 37, 2546-59.
- Kleiber, C. and S. Kotz (2003). Statistical Size Distributions in Economics and Actuarial Sciences. New York: Wiley.
- Loretan, M. and P. Phillips (1994). Testing the covariance structure of heavy-tailed time series. Journal of Empirical Finance, 1, 211-48.
- Maronna, R., Martin, D. and Yohai, V. (2006). Robust Statistics: Theory and Methods. John Wiley & Sons Ltd.
- McDonald J. B. and Y. J. Xu (1995). A generalization of the beta distribution with applications. Journal of Econometrics 66: 133-152.
- McNeil, A.J., Frey, R. and P. Embrechts (2005). Quantitative Risk Management. Princeton University Press, Princeton.
- Mikosch, T. and C. Starica (2000). Limit theory for the sample autocorrelations and extremes of a GARCH(1,1) process. *Annals of Staistics*, 28, 1427-51.
- Muler, N. and V. J. Yohai (2008). Robust estimates for GARCH models. Journal of Statistical Planning and Inference 138, 2918-2940.
- Nelson, D.B. (1991). Conditional heteroskedasticity in asset returns: a new approach. *Econometrica* 59, 347-370.
- Prentice, R. L. (1975). Discrimination among some parametric models. *Biometrika*, 62, 607-614.

- Resnick, S. and C. Starica (1995). Consistency of Hill's Estimator for Dependent Data. Journal of Applied Probability, 32, 139-167.
- Wang, K-L, Fawson, C., Barrett, C.B. and J. B. McDonald.(2001). Journal of Applied Econometrics, 16, 521-536.

- N. 930 Uncertainty and heterogeneity in factor models forecasting, by Matteo Luciani and Libero Monteforte (September 2013).
- N. 931 *Economic insecurity and fertility intentions: the case of Italy*, by Francesca Modena, Concetta Rondinelli and Fabio Sabatini (September 2013).
- N. 932 The role of regulation on entry: evidence from the Italian provinces, by Francesco Bripi (September 2013).
- N. 933 *The management of interest rate risk during the crisis: evidence from Italian banks*, by Lucia Esposito, Andrea Nobili and Tiziano Ropele (September 2013).
- N. 934 *Central bank and government in a speculative attack model*, by Giuseppe Cappelletti and Lucia Esposito (September 2013).
- N. 935 *Ita-coin: a new coincident indicator for the Italian economy*, by Valentina Aprigliano and Lorenzo Bencivelli (October 2013).
- N. 936 *The Italian financial cycle: 1861-2011*, by Riccardo De Bonis and Andrea Silvestrini (October 2013).
- N. 937 The effect of tax enforcement on tax morale, by Antonio Filippin, Carlo V. Fiorio and Eliana Viviano (October 2013).
- N. 938 *Tax deferral and mutual fund inflows: evidence from a quasi-natural experiment*, by Giuseppe Cappelletti, Giovanni Guazzarotti and Pietro Tommasino (November 2013).
- N. 939 Shadow banks and macroeconomic instability, by Roland Meeks, Benjamin Nelson and Piergiorgio Alessandri (November 2013).
- N. 940 Heterogeneous firms and credit frictions: a general equilibrium analysis of market entry decisions, by Sara Formai (November 2013).
- N. 941 The trend-cycle decomposition of output and the Phillips curve: Bayesian estimates for Italy, by Fabio Busetti and Michele Caivano (November 2013).
- N. 942 Supply tightening or lack of demand? An analysis of credit developments during the Lehman Brothers and the sovereign debt crises, by Paolo Del Giovane, Andrea Nobili and Federico Maria Signoretti (November 2013).
- N. 943 Sovereign risk, monetary policy and fiscal multipliers: a structural model-based assessment, by Alberto Locarno, Alessandro Notarpietro and Massimiliano Pisani (November 2013).
- N. 944 *Calibrating the Italian smile with time-varying volatility and heavy-tailed models*, by Michele Leonardo Bianchi, Frank J. Fabozzi and Svetlozar T. Rachev (January 2014).
- N. 945 Simple banking: profitability and the yield curve, by Piergiorgio Alessandri and Benjamin Nelson (January 2014).
- N. 946 Information acquisition and learning from prices over the business cycle, by Taneli Mäkinen and Björn Ohl (January 2014).
- N. 947 *Time series models with an EGB2 conditional distribution*, by Michele Caivano and Andrew Harvey (January 2014).
- N. 948 Trade and finance: is there more than just 'trade finance'? Evidence from matched bank-firm data, by Silvia Del Prete and Stefano Federico (January 2014).
- N. 949 *Natural disasters, growth and institutions: a tale of two earthquakes*, by Guglielmo Barone and Sauro Mocetti (January 2014).
- N. 950 *The cost of firms' debt financing and the global financial crisis*, by Daniele Pianeselli and Andrea Zaghini (February 2014).
- N. 951 On bank credit risk: systemic or bank-specific? Evidence from the US and UK, by Junye Li and Gabriele Zinna (February 2014).
- N. 952 School cheating and social capital, by Marco Paccagnella and Paolo Sestito (February 2014).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- A. PRATI and M. SBRACIA, *Uncertainty and currency crises: evidence from survey data*, Journal of Monetary Economics, v, 57, 6, pp. 668-681, **TD No. 446 (July 2002).**
- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, v. 42, 19-21, pp. 2399-2415, **TD No. 458 (December 2002).**
- S. MAGRI, *Debt maturity choice of nonpublic Italian firms*, Journal of Money, Credit, and Banking, v.42, 2-3, pp. 443-463, **TD No. 574 (January 2006).**
- G. DE BLASIO and G. NUZZO, *Historical traditions of civicness and local economic development*, Journal of Regional Science, v. 50, 4, pp. 833-857, **TD No. 591 (May 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, v. 62, 2, pp. 374-394, **TD No. 598 (September 2006).**
- S. NERI and A. NOBILI, *The transmission of US monetary policy to the euro area,* International Finance, v. 13, 1, pp. 55-78, **TD No. 606 (December 2006).**
- F. ALTISSIMO, R. CRISTADORO, M. FORNI, M. LIPPI and G. VERONESE, *New Eurocoin: Tracking Economic Growth in Real Time*, Review of Economics and Statistics, v. 92, 4, pp. 1024-1034, **TD No. 631 (June 2007).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and taxation*, Journal of Banking and Finance, v. 34, 11, pp. 2801-2810, **TD No. 649** (November 2007).
- L. GAMBACORTA and C. ROSSI, *Modelling bank lending in the euro area: a nonlinear approach*, Applied Financial Economics, v. 20, 14, pp. 1099-1112 ,**TD No. 650** (November 2007).
- M. IACOVIELLO and S. NERI, *Housing market spillovers: evidence from an estimated DSGE model,* American Economic Journal: Macroeconomics, v. 2, 2, pp. 125-164, **TD No. 659 (January 2008).**
- F. BALASSONE, F. MAURA and S. ZOTTERI, *Cyclical asymmetry in fiscal variables in the EU*, Empirica, **TD** No. 671, v. 37, 4, pp. 381-402 (June 2008).
- F. D'AMURI, GIANMARCO I.P. OTTAVIANO and G. PERI, *The labor market impact of immigration on the western german labor market in the 1990s*, European Economic Review, v. 54, 4, pp. 550-570, **TD No. 687 (August 2008).**
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, v. 89, 1, pp. 173-190, **TD No. 688 (September 2008).**
- S. NOBILI and G. PALAZZO, *Explaining and forecasting bond risk premiums*, Financial Analysts Journal, v. 66, 4, pp. 67-82, **TD No. 689 (September 2008).**
- A. B. ATKINSON and A. BRANDOLINI, *On analysing the world distribution of income*, World Bank Economic Review, v. 24, 1, pp. 1-37, **TD No. 701 (January 2009).**
- R. CAPPARIELLO and R. ZIZZA, Dropping the Books and Working Off the Books, Labour, v. 24, 2, pp. 139-162, **TD No. 702 (January 2009).**
- C. NICOLETTI and C. RONDINELLI, *The (mis)specification of discrete duration models with unobserved heterogeneity: a Monte Carlo study*, Journal of Econometrics, v. 159, 1, pp. 1-13, **TD No. 705** (March 2009).
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector: the case of Italy*, Macroeconomic Dynamics, v. 14, 5, pp. 677-708, **TD No. 706** (March 2009).
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS-IBÁÑEZ, *Bank risk and monetary policy*, Journal of Financial Stability, v. 6, 3, pp. 121-129, **TD No. 712** (May 2009).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Dynamic macroeconomic effects of public capital: evidence from regional Italian data, Giornale degli economisti e annali di economia, v. 69, 1, pp. 29-66, TD No. 733 (November 2009).
- F. COLUMBA, L. GAMBACORTA and P. E. MISTRULLI, *Mutual Guarantee institutions and small business finance*, Journal of Financial Stability, v. 6, 1, pp. 45-54, **TD No. 735** (November 2009).
- A. GERALI, S. NERI, L. SESSA and F. M. SIGNORETTI, *Credit and banking in a DSGE model of the Euro Area,* Journal of Money, Credit and Banking, v. 42, 6, pp. 107-141, **TD No. 740** (January 2010).
- M. AFFINITO and E. TAGLIAFERRI, *Why do (or did?) banks securitize their loans? Evidence from Italy*, Journal of Financial Stability, v. 6, 4, pp. 189-202, **TD No. 741 (January 2010).**
- S. FEDERICO, Outsourcing versus integration at home or abroad and firm heterogeneity, Empirica, v. 37, 1, pp. 47-63, **TD No. 742** (February 2010).

- V. DI GIACINTO, *On vector autoregressive modeling in space and time*, Journal of Geographical Systems, v. 12, 2, pp. 125-154, **TD No. 746 (February 2010).**
- L. FORNI, A. GERALI and M. PISANI, *The macroeconomics of fiscal consolidations in euro area countries,* Journal of Economic Dynamics and Control, v. 34, 9, pp. 1791-1812, **TD No. 747** (March 2010).
- S. MOCETTI and C. PORELLO, *How does immigration affect native internal mobility? new evidence from Italy*, Regional Science and Urban Economics, v. 40, 6, pp. 427-439, **TD No. 748 (March 2010)**.
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap spread changes before and during the subprime financial turmoil, Journal of Current Issues in Finance, Business and Economics, v. 3, 4, pp., **TD No. 749** (March 2010).
- P. CIPOLLONE, P. MONTANARO and P. SESTITO, Value-added measures in Italian high schools: problems and findings, Giornale degli economisti e annali di economia, v. 69, 2, pp. 81-114, TD No. 754 (March 2010).
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, Asset-based measurement of poverty, Journal of Policy Analysis and Management, v. 29, 2, pp. 267-284, **TD No. 755** (March 2010).
- G. CAPPELLETTI, A Note on rationalizability and restrictions on beliefs, The B.E. Journal of Theoretical Economics, v. 10, 1, pp. 1-11, **TD No. 757** (April 2010).
- S. DI ADDARIO and D. VURI, Entrepreneurship and market size. the case of young college graduates in *Italy*, Labour Economics, v. 17, 5, pp. 848-858, **TD No. 775 (September 2010).**
- A. CALZA and A. ZAGHINI, *Sectoral money demand and the great disinflation in the US*, Journal of Money, Credit, and Banking, v. 42, 8, pp. 1663-1678, **TD No. 785 (January 2011).**

2011

- S. DI ADDARIO, *Job search in thick markets*, Journal of Urban Economics, v. 69, 3, pp. 303-318, **TD No.** 605 (December 2006).
- F. SCHIVARDI and E. VIVIANO, *Entry barriers in retail trade*, Economic Journal, v. 121, 551, pp. 145-170, **TD** No. 616 (February 2007).
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing excess liquidity in the Euro Area: the role of sectoral distribution of money, Applied Economics, v. 43, 23, pp. 3213-3230, **TD No. 627** (April 2007).
- P. E. MISTRULLI, Assessing financial contagion in the interbank market: maximum entropy versus observed interbank lending patterns, Journal of Banking & Finance, v. 35, 5, pp. 1114-1127, TD No. 641 (September 2007).
- E. CIAPANNA, *Directed matching with endogenous markov probability: clients or competitors?*, The RAND Journal of Economics, v. 42, 1, pp. 92-120, **TD No. 665 (April 2008).**
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, v. 78, 311, pp. 480-500, **TD No. 673 (June 2008).**
- V. DI GIACINTO e M. PAGNINI, Local and global agglomeration patterns: two econometrics-based indicators, Regional Science and Urban Economics, v. 41, 3, pp. 266-280, **TD No. 674 (June 2008)**.
- G. BARONE and F. CINGANO, Service regulation and growth: evidence from OECD countries, Economic Journal, v. 121, 555, pp. 931-957, TD No. 675 (June 2008).
- P. SESTITO and E. VIVIANO, *Reservation wages: explaining some puzzling regional patterns*, Labour, v. 25, 1, pp. 63-88, **TD No. 696 (December 2008).**
- R. GIORDANO and P. TOMMASINO, *What determines debt intolerance? The role of political and monetary institutions*, European Journal of Political Economy, v. 27, 3, pp. 471-484, **TD No. 700 (January 2009).**
- P. ANGELINI, A. NOBILI e C. PICILLO, *The interbank market after August 2007: What has changed, and why?*, Journal of Money, Credit and Banking, v. 43, 5, pp. 923-958, **TD No. 731 (October 2009).**
- G. BARONE and S. MOCETTI, *Tax morale and public spending inefficiency*, International Tax and Public Finance, v. 18, 6, pp. 724-49, **TD No. 732 (November 2009).**
- L. FORNI, A. GERALI and M. PISANI, *The Macroeconomics of Fiscal Consolidation in a Monetary Union:* the Case of Italy, in Luigi Paganetto (ed.), Recovery after the crisis. Perspectives and policies, VDM Verlag Dr. Muller, **TD No. 747 (March 2010).**
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap changes before and during the subprime financial turmoil, in Barbara L. Campos and Janet P. Wilkins (eds.), The Financial Crisis: Issues in Business, Finance and Global Economics, New York, Nova Science Publishers, Inc., TD No. 749 (March 2010).

- A. LEVY and A. ZAGHINI, *The pricing of government guaranteed bank bonds*, Banks and Bank Systems, v. 6, 3, pp. 16-24, **TD No. 753 (March 2010).**
- G. BARONE, R. FELICI and M. PAGNINI, *Switching costs in local credit markets,* International Journal of Industrial Organization, v. 29, 6, pp. 694-704, **TD No. 760 (June 2010).**
- G. BARBIERI, C. ROSSETTI e P. SESTITO, The determinants of teacher mobility: evidence using Italian teachers' transfer applications, Economics of Education Review, v. 30, 6, pp. 1430-1444, TD No. 761 (marzo 2010).
- G. GRANDE and I. VISCO, A public guarantee of a minimum return to defined contribution pension scheme members, The Journal of Risk, v. 13, 3, pp. 3-43, **TD No. 762 (June 2010).**
- P. DEL GIOVANE, G. ERAMO and A. NOBILI, *Disentangling demand and supply in credit developments: a survey-based analysis for Italy*, Journal of Banking and Finance, v. 35, 10, pp. 2719-2732, **TD No.** 764 (June 2010).
- G. BARONE and S. MOCETTI, With a little help from abroad: the effect of low-skilled immigration on the female labour supply, Labour Economics, v. 18, 5, pp. 664-675, **TD No. 766 (July 2010).**
- S. FEDERICO and A. FELETTIGH, *Measuring the price elasticity of import demand in the destination markets of italian exports*, Economia e Politica Industriale, v. 38, 1, pp. 127-162, **TD No. 776 (October 2010).**
- S. MAGRI and R. PICO, *The rise of risk-based pricing of mortgage interest rates in Italy*, Journal of Banking and Finance, v. 35, 5, pp. 1277-1290, **TD No. 778 (October 2010).**
- M. TABOGA, Under/over-valuation of the stock market and cyclically adjusted earnings, International Finance, v. 14, 1, pp. 135-164, **TD No. 780 (December 2010).**
- S. NERI, *Housing, consumption and monetary policy: how different are the U.S. and the Euro area?*, Journal of Banking and Finance, v.35, 11, pp. 3019-3041, **TD No. 807** (April 2011).
- V. CUCINIELLO, *The welfare effect of foreign monetary conservatism with non-atomistic wage setters*, Journal of Money, Credit and Banking, v. 43, 8, pp. 1719-1734, **TD No. 810 (June 2011).**
- A. CALZA and A. ZAGHINI, welfare costs of inflation and the circulation of US currency abroad, The B.E. Journal of Macroeconomics, v. 11, 1, Art. 12, **TD No. 812 (June 2011).**
- I. FAIELLA, *La spesa energetica delle famiglie italiane*, Energia, v. 32, 4, pp. 40-46, **TD No. 822 (September 2011).**
- R. DE BONIS and A. SILVESTRINI, *The effects of financial and real wealth on consumption: new evidence from* OECD countries, Applied Financial Economics, v. 21, 5, pp. 409–425, **TD No. 837 (November 2011).**
- F. CAPRIOLI, P. RIZZA and P. TOMMASINO, *Optimal fiscal policy when agents fear government default*, Revue Economique, v. 62, 6, pp. 1031-1043, **TD No. 859** (March 2012).

2012

- F. CINGANO and A. ROSOLIA, *People I know: job search and social networks*, Journal of Labor Economics, v. 30, 2, pp. 291-332, **TD No. 600 (September 2006).**
- G. GOBBI and R. ZIZZA, Does the underground economy hold back financial deepening? Evidence from the italian credit market, Economia Marche, Review of Regional Studies, v. 31, 1, pp. 1-29, TD No. 646 (November 2006).
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, v. 20, 2, pp. 189-209, **TD No. 691 (September 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, v. 10, 6, pp. 1318–1347, **TD No. 698 (December 2008).**
- M. PERICOLI and M. TABOGA, *Bond risk premia, macroeconomic fundamentals and the exchange rate,* International Review of Economics and Finance, v. 22, 1, pp. 42-65, **TD No. 699 (January 2009).**
- F. LIPPI and A. NOBILI, *Oil and the macroeconomy: a quantitative structural analysis*, Journal of European Economic Association, v. 10, 5, pp. 1059-1083, **TD No. 704** (March 2009).
- G. ASCARI and T. ROPELE, *Disinflation in a DSGE perspective: sacrifice ratio or welfare gain ratio?*, Journal of Economic Dynamics and Control, v. 36, 2, pp. 169-182, **TD No. 736 (January 2010).**
- S. FEDERICO, *Headquarter intensity and the choice between outsourcing versus integration at home or abroad*, Industrial and Corporate Chang, v. 21, 6, pp. 1337-1358, **TD No. 742 (February 2010).**
- I. BUONO and G. LALANNE, *The effect of the Uruguay Round on the intensive and extensive margins of trade*, Journal of International Economics, v. 86, 2, pp. 269-283, **TD No. 743 (February 2010).**

- S. GOMES, P. JACQUINOT and M. PISANI, The EAGLE. A model for policy analysis of macroeconomic interdependence in the euro area, Economic Modelling, v. 29, 5, pp. 1686-1714, TD No. 770 (July 2010).
- A. ACCETTURO and G. DE BLASIO, Policies for local development: an evaluation of Italy's "Patti Territoriali", Regional Science and Urban Economics, v. 42, 1-2, pp. 15-26, TD No. 789 (January 2006).
- F. BUSETTI and S. DI SANZO, *Bootstrap LR tests of stationarity, common trends and cointegration,* Journal of Statistical Computation and Simulation, v. 82, 9, pp. 1343-1355, **TD No. 799 (March 2006).**
- S. NERI and T. ROPELE, *Imperfect information, real-time data and monetary policy in the Euro area,* The Economic Journal, v. 122, 561, pp. 651-674, **TD No. 802 (March 2011).**
- G. CAPPELLETTI, G. GUAZZAROTTI and P. TOMMASINO, *What determines annuity demand at retirement?*, The Geneva Papers on Risk and Insurance – Issues and Practice, pp. 1-26, **TD No. 805 (April 2011).**
- A. ANZUINI and F. FORNARI, *Macroeconomic determinants of carry trade activity*, Review of International Economics, v. 20, 3, pp. 468-488, **TD No. 817 (September 2011).**
- M. AFFINITO, *Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy*, Journal of Banking and Finance, v. 36, 12, pp. 3163-3184, **TD No. 826 (October 2011).**
- R. CRISTADORO and D. MARCONI, *Household savings in China*, Journal of Chinese Economic and Business Studies, v. 10, 3, pp. 275-299, **TD No. 838 (November 2011).**
- P. GUERRIERI and F. VERGARA CAFFARELLI, Trade Openness and International Fragmentation of Production in the European Union: The New Divide?, Review of International Economics, v. 20, 3, pp. 535-551, TD No. 855 (February 2012).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Network effects of public transposrt infrastructure: evidence on Italian regions, Papers in Regional Science, v. 91, 3, pp. 515-541, TD No. 869 (July 2012).
- A. FILIPPIN and M. PACCAGNELLA, *Family background, self-confidence and economic outcomes,* Economics of Education Review, v. 31, 5, pp. 824-834, **TD No. 875 (July 2012).**

2013

- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, v. 11, 2, pp. 433-465, **TD No. 709 (May 2009).**
- F. BUSETTI and J. MARCUCCI, *Comparing forecast accuracy: a Monte Carlo investigation*, International Journal of Forecasting, v. 29, 1, pp. 13-27, **TD No. 723 (September 2009).**
- D. DOTTORI, S. I-LING and F. ESTEVAN, *Reshaping the schooling system: The role of immigration*, Journal of Economic Theory, v. 148, 5, pp. 2124-2149, **TD No. 726 (October 2009).**
- A. FINICELLI, P. PAGANO and M. SBRACIA, *Ricardian Selection*, Journal of International Economics, v. 89, 1, pp. 96-109, **TD No. 728 (October 2009).**
- L. MONTEFORTE and G. MORETTI, *Real-time forecasts of inflation: the role of financial variables*, Journal of Forecasting, v. 32, 1, pp. 51-61, **TD No. 767 (July 2010).**
- E. GAIOTTI, Credit availablility and investment: lessons from the "Great Recession", European Economic Review, v. 59, pp. 212-227, TD No. 793 (February 2011).
- A. ACCETTURO e L. INFANTE, Skills or Culture? An analysis of the decision to work by immigrant women in Italy, IZA Journal of Migration, v. 2, 2, pp. 1-21, **TD No. 815 (July 2011).**
- A. DE SOCIO, *Squeezing liquidity in a "lemons market" or asking liquidity "on tap"*, Journal of Banking and Finance, v. 27, 5, pp. 1340-1358, **TD No. 819 (September 2011).**
- M. FRANCESE and R. MARZIA, is there Room for containing healthcare costs? An analysis of regional spending differentials in Italy, The European Journal of Health Economics (DOI 10.1007/s10198-013-0457-4), TD No. 828 (October 2011).
- G. BARONE and G. DE BLASIO, *Electoral rules and voter turnout*, International Review of Law and Economics, v. 36, 1, pp. 25-35, **TD No. 833 (November 2011).**
- E. GENNARI and G. MESSINA, How sticky are local expenditures in Italy? Assessing the relevance of the flypaper effect through municipal data, International Tax and Public Finance (DOI: 10.1007/s10797-013-9269-9), TD No. 844 (January 2012).
- A. ANZUINI, M. J. LOMBARDI and P. PAGANO, *The impact of monetary policy shocks on commodity prices*, International Journal of Central Banking, v. 9, 3, pp. 119-144, **TD No. 851 (February 2012).**

S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics (DOI: 10.1111/obes.12023), **TD No. 879 (September 2012).**

FORTHCOMING

- A. MERCATANTI, A likelihood-based analysis for relaxing the exclusion restriction in randomized experiments with imperfect compliance, Australian and New Zealand Journal of Statistics, TD No. 683 (August 2008).
- M. TABOGA, *The riskiness of corporate bonds*, Journal of Money, Credit and Banking, **TD No. 730** (October 2009).
- F. D'AMURI, *Gli effetti della legge 133/2008 sulle assenze per malattia nel settore pubblico*, Rivista di Politica Economica, **TD No. 787 (January 2011).**
- E. COCOZZA and P. PISELLI, Testing for east-west contagion in the European banking sector during the financial crisis, in R. Matoušek; D. Stavárek (eds.), Financial Integration in the European Union, Taylor & Francis, TD No. 790 (February 2011).
- R. BRONZINI and E. IACHINI, Are incentives for R&D effective? Evidence from a regression discontinuity approach, American Economic Journal : Economic Policy, **TD No. 791 (February 2011).**
- F. NUCCI and M. RIGGI, *Performance pay and changes in U.S. labor market dynamics*, Journal of Economic Dynamics and Control, **TD No. 800 (March 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, **TD No. 835 (November 2011).**
- F. D'AMURI and G. PERI, Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession, Journal of the European Economic Association, TD No. 886 (October 2012).
- R. DE BONIS and A. SILVESTRINI, *The Italian financial cycle: 1861-2011*, Cliometrica, **TD No. 936** (October 2013).